Author's response to reviews

Title: Validity of Simpson-Angus Scale (SAS) in a naturalistic schizophrenia population

Authors:

Sven Janno (svenjanno@hot.ee)
Matti M Holi (matti.holi@hus.fi)
Katinka Tuisku (katinka.tuisku@ttl.fi)
Kristian Wahlbeck (kristian.wahlbeck@stakes.fi)

Version: 4 Date: 6 March 2005

Author's response to reviews:

Dear editor-in-chief,

We are very happy that you accepted our manuscript for publishing. We would like to publish it in BMC Neurology.

Thank you for the valuable referee comments on our manuscript.

Reviewer 1 (Caligiuri):

Major Compulsory Revisions:
1) We added the sentence about nonsignificant differences of some items (p 5, lines 26-28).
2) We added a sentence of meaning of AUC values of ROC curve in Results part (p 6, lines 10-11).

We made the requested formatting changes.
We declare that we have no competing interests.

On behalf of the Authors,

Sven Janno